Author’s response to reviews

Title: Who drinks sugar sweetened beverages and juice? An Australian population study of behaviour, awareness and attitudes

Authors:

Caroline Miller (Caroline.Miller@sahmri.com)
Melanie Wakefield (Melanie.Wakefield@cancervic.org.au)
Annette Braunack-Mayer (abmayer@uow.edu.au)
David Roder (David.Roder@unisa.edu.au)
Kerin O'Dea (kerin.odea@unisa.edu.au)
Kerry Ettridge (kerry.ettridge@sahmri.com)
Joanne Dono (jo.dono@sahmri.com)

Version: 1 Date: 06 May 2018

Author’s response to reviews:

Dear Amelia and Francesca,

The manuscript has been updated as requested.

Kind regards,

A/Prof Caroline Miller